Author's response to reviews

Title: Glucagonoma syndrome: a case report

Authors:

Pablo Granero Castro (pgranerocastro@aecirujanos.es)
Alberto Miyar de León (albertmiyar@yahoo.es)
Jose Granero Trancón (jose.granero@aecirujanos.es)
Paloma Álvarez Martínez (dra.palialvarez@gmail.com)
Jose A Álvarez Pérez (josealvar@telecable.es)
Jose C Fernández Fernández (blendio@terra.es)
Carmen M García Bernardo (cmgbernardo2@hotmail.com)
Luis Barneo Serra (lbarneo@uniovi.es)
Juan J González González (jjglezglez@gmail.com)

Version: 7 Date: 5 May 2011

Author's response to reviews: see over
May 5, 2011.

The Biomed Central Editorial Team


Thank you for consideration of our manuscript for publication in your journal.

Attending to the editorial team considerations, the following changes have been made in the manuscript:

- Please upload the figures as separate files via the online submission system:
  ✓ Done.

- Please include a figure title and legend section after the reference list:
  ✓ Done. Figure legend included in page 12 after reference list.

PD: As other Biomed manuscripts published, the reference item should be “Granero Castro et al”.